Author's response to reviews

Title: Identification of Cassia tora Trypsin inhibitor active against proteases of Aspergillus flavus and Bacillus sp.

Authors:

Vinayak R Tripathi (trip.vins@gmail.com)
Shailendra Kumar (shailendra.microbio@gmail.com)
Satyendra K Garg (sk_garg001@yahoo.com)

Version: 2 Date: 9 December 2010

Author's response to reviews: see over
To,

The Editor-in-Chief

BMC Complementary and Alternative Medicine,

Dear Sir,

Please find attached herewith a revised manuscript entitled ‘Identification of *Cassia tora* Trypsin inhibitor active against proteases of *Aspergillus flavus* and *Bacillus* sp. The manuscript was previously submitted to the BMC Plant Biology (MS: 1702705549475305). However, as per the message from Editorial Administrator, the manuscript is now modified and being submitted to BMC Complementary and Alternative Medicine. The authors agree that the manuscript is suitable for publication in BMC Complementary and Alternative Medicine as it explores presence of hitherto unreported protease inhibitor (PI) in *Cassia tora*, a medicinal plant prescribed in skin and gastrointestinal disorders. The activity of inhibitor is established against trypsin, bacterial and fungal proteases. As per the required format of the journal, the manuscript has been aptly modified. It is an original research work. All authors have approved the manuscript for final submission. The manuscript is not being considered for publication, in whole or in part, in another journal, book, and conference proceedings etc.

Merry Christmas and with season’s greetings,

Sincerely

S. K. Garg